

Tilburg University

Saving, openness, and growth

Hoogstrate, A.J.; Osang, T.

Publication date: 1998

Link to publication in Tilburg University Research Portal

Citation for published version (APA): Hoogstrate, A. J., & Osang, T. (1998). Saving, openness, and growth. (CentER Discussion Paper; Vol. 1998-47). CentER.

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
 You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal

Take down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.



Center for Economic Research

No. 9847

SAVING, OPENNESS, AND GROWTH

By André J. Hoogstrate and Thomas Osang

May 1998

ISSN 0924-7815

Saving, Openness, and Growth

André J. Hoogstrate,^{*†} Tilburg University, Tilburg

Thomas Osang[‡] Southern Methodist University, Dallas

April, 1998

Keywords: endogenous threshold estimation, non-linear VAR, impulse response analysis, international trade. JEL classification: C1, F4.

^{*}We would like to thank Herman Bierens, Jean Pierre Urbain, Gerard Pfann, and Paul Macquarie as well as seminar participants at the Mid-West International Economics Meetings, Fall 97, and the ASSA Meetings, Chicago, 1998, for helpful comments and suggestions.

[†]Tilburg University, Faculty of Economics and Business Administration, P.O. Box 90153, 5000 LE Tilburg, The Netherlands. Tel: +31-13-466-2031, Fax: +31-13-466-3280, E-mail: a.j.hoogstrate@kub.nl. This research was sponsored by the Economics Research Foundation, which is part of the Netherlands Organization for Scientific Research (NWO).

[‡]Southern Methodist University, Department of Economics, Dallas, TX 75275, Tel: (214) 768-4398, Fax: (214) 768-1821, E-mail: tosang@mail.smu.edu. This research was supported by a grant from the University Research Council of SMU.

Abstract

Using a country-level panel data set, we investigate whether a country's saving rate matters for the relationship between GDP growth and openness to trade. We first derive a new LM-type statistic to test for the existence of an endogenous threshold separating high from low savings regimes. Once existence is established, estimation of threshold VAR models shows that for countries in the high savings regime, openness has a positive effect on growth, while it has no effect for countries in the low savings regime. Furthermore, there are striking differences in the impulse response functions between the two regimes. In the high savings regime, a positive shock in openness leads to higher GDP growth in all subsequent years, while in the low savings regime, the effect on GDP growth is substantially smaller in size and becomes insignificant after a few periods.

1 Introduction

This paper brings together two important parts of the empirical literature on the determinants of economic growth, namely growth and openness to trade on the one hand and growth and saving on the other. By now, there exists a large number of empirical studies on the relationship between growth and various measures of openness to trade (See, for example, Harrison [24], Levine and Renelt [29], Edwards [14], [15], Jorgenson and Ho [26], Balassa [5], [6], and Quah and Rauch [35]). Surprisingly, many of these studies find a rather weak relationship between openness to trade and growth. The estimated coefficient on openness is often statistically insignificant or even has the wrong sign. In Anne Harrison's study, for example, only two of the six measures of openness are significant at the five percent level, while the sign of the trade share in GDP is negative though not significant (Table 6, p. 434). Furthermore, some studies find the estimated coefficient on openness to be sensitive to changes in the econometric model or data set. Levine and Renelt report that "after controlling for the share of investment in GDP we cannot find an independent and robust relationship between any trade ...indicator and growth" (p. 954).

In contrast to the results for the empirical relationship between openness and growth, the empirical literature provides solid evidence on the relationship between growth and saving (See, for example, Maddison [30], Carroll and Weil [11], Bosworth [10], Gupta and Islam [19]). These studies find that countries with higher saving rates have significantly higher growth rates, a result that sensitivity tests show to be fairly robust (Levine and Renelt [29], p. 946)¹.

Interestingly, none of the studies cited above examines the potentially nonlinear relationship between the GDP growth, openness, and saving. Such a nonlinear link has been the focus of recent theoretical models from the literature on trade and endogenous growth. These models indicate that certain model parameters linked to the consumption and savings behavior of households may play a key role in the interaction between growth and openness (Feenstra [16], Osang and Pereira [31], [32]). Feenstra shows that trade can increase or decrease a country's growth rate depending on the value of

¹Levine and Renelt show that the relationship between the investment share in GDP and GDP growth is strong and robust. Since it is well known that investment and saving rates are highly correlated within countries, we interpret their findings as indirect evidence for a robust link between saving and GDP growth.

the *instantaneous* elasticity of substitution, while Osang and Pereira point out that there exists a threshold level of the *intertemporal* elasticity of substitution that separates growth-enhancing from growth-reducing regimes of increased openness to trade². Since the intertemporal elasticity of substitution is a key parameter for saving in the endogenous growth literature, it seems naturally to test the hypothesis that the relationship between openness and growth changes once the saving rate exceeds a certain endogenously determined threshold level. If the hypothesis cannot be rejected by the data, one could argue that the weak link between openness and growth found in the literature may simply be the result of a misspecified model.

In order to test the above hypothesis, we first test for the existence of a threshold saving rate separating high from low saving regimes. We then use a dynamic model and simultaneously estimate the benchmark value of the savings rate and the other parameters of the model using data from 58 countries for the period from 1960 to 1987. The main findings of the paper are as follows. First, using different threshold test statistics we show the existence of an endogenous threshold separating high from low savings regimes. Second, we find that saving rates indeed matter for the link between openness and growth. For countries in the high savings regime, openness to trade has a positive effect on growth, while it has no effect on growth for countries in the low savings regime. Third, there are striking differences in the impulse response functions between the two regimes. In the high savings regime, a positive shock in openness leads to higher GDP growth in all subsequent years, while in the low savings regime the effect on GDP growth is substantially smaller in size and becomes insignificant after a few periods. Interestingly, a positive shock in openness leads to a further decline in savings in the low savings regime.

The paper extends the existing literature in three ways. First, we explicitly estimate the non-linear relationship between saving rates, openness to trade, and growth, while the standard growth literature cited above uses a linear relationship at best. Second, by employing a dynamic model instead of the widely used cross-section analysis, we allow the effect of saving on

²A possible mechanism through which openness affects growth is as follows (see Osang and Pereira [31], [32] for details). International differences in preferences and/or technologies lead to different steady state growth rates across countries. Assuming balanced trade and complete specialization, increasing the volume of trade in both countries (e.g. due to lower trade barriers or changes in consumer preferences) induces changes in the terms of trade. In this situation it is most likely that the country with the weak attitude toward saving will experience an improvement in its terms of trade and, in turn, a decline in output growth, while the country with the strong saving performance will experience the opposite effects.

the relationship between openness and growth to work in the cross section (across countries) as well as the time dimension. This is important since the cross section approach is only justifiable in very specific cases as shown by Harrison [24]. Third, estimating an endogenous threshold in a dynamic model raises some interesting questions concerning the underlying econometric theory since the estimation model is not well defined in the case of a non-existent threshold level. To this regard we introduce a new test statistic which allows us to test for the existence of threshold saving level. The test is based on results derived by Bierens and Ploberger [9] and can be considered as an alternative to tests by Andrews et al. [1] and Andrews and Ploberger [2]. One advantage of the new test is that it is easier to calculate than, for example, the threshold test suggested by Hansen [21], [22].

The remainder of the paper is organized as follows. Section 2 describes the estimated model in detail. We discuss the econometric methodology necessary for estimation and testing of thresholds in panel data sets in section 3. Section 4 describes the data, while section 5 contains the empirical results. Section 6 concludes the paper.

2 The Empirical Model

The recent theoretical literature on growth in open economies suggests that GDP growth mainly depends on the following parameters³:

- Technology parameters such as total factor productivity, A, and scale elasticities, α .
- Taste parameters such as the intertemporal elasticity of substitution, σ , and the discount rate, ρ .
- Trade policy parameters, τ , measuring a country's tariff and non-tariff barriers to trade.

The relation can thus be written as

$$growth = f(A, \alpha, \sigma, \rho, \tau).$$
(1)

Unfortunately, in most cases we lack data that directly measure these parameters, especially over time. It is thus common in the literature to approximate these parameters with data that are available both across countries and time.

³See, for example, Lee [27], Rivera-Batiz and Romer [36], and Turnovsky [38].

Taste parameters reflect a country's willingness to postpone current consumption and thus determine the domestic supply of financial capital. Advanced production technologies are intensive in both physical and human capital and correspond to high levels of total factor productivity. Advanced production technologies are thus a major factor behind the domestic demand for financial capital. Given the fact that international capital is not sufficiently mobile (see Gordon and Bovenberg [18] for a recent investigation of this well-known puzzle) it seems reasonable to use the domestic saving rate of a country as a proxy for both sufficient supply and adequate demand in the market for financial capital. Trade policy parameters can be approximated either directly through some index of trade liberalization using country sources on trade barriers (see, for example, Thomas et al [37]) or indirectly through measures such as the blackmarket premium in the currency market or an index measuring price distortions for consumption goods⁴. A widely used indirect measure for trade barriers is the ratio of exports plus imports to GDP. This proxy has the advantage that it is available for many countries for at least three decades. It is also relatively free of different definitions and data collection techniques between countries. Furthermore, Harrison [24] shows that it has the highest correlation coefficient with trade reform compared to all other indirect openness measures (Table 2, p. 429). Its most severe disadvantage is that its value does not depend on trade barriers alone but also on country size or foreign direct investment. Nevertheless, we will use the share of trade in GDP as our proxy for a country's openness to trade. Finally, it is common in the literature to assume that the technology parameters can be incorporated in the function f. This leaves us with the following estimable model:

$$growth = f(s, o) \tag{2}$$

where s indicates the saving rate and o indicates the level of openness. Note that (2) is also a common result in the endogenous growth literature where output growth can be expressed as a function of the (endogenous) savings rate of the economy. Since (2) expresses an equilibrium condition which may or may not hold in reality due to exogenous shocks, it is important to include dynamics in our empirical model. For this reason and because the functional form of f is unknown we use an unrestricted linear first-order VAR model as an approximation of (2),

$$X_{i,t} = c + A_1 X_{i,t-1} + \varepsilon_{i,t},\tag{3}$$

⁴See Harrison [24] for other indirect measures of trade barriers as well as a detailed dicussion of measurement problems associated with both direct and indirect proxies of trade barriers.

with $X_{i,t} = (O_{i,t}, S_{i,t}, \Delta Y_{i,t})'$ where

- $\Delta Y_{i,t}$ denotes GDP growth of country *i* at time *t*, $(\Delta Y_{i,t} = log(Y_{i,t}) log(Y_{i,t-1}))$
- $S_{i,t}$ denotes the gross domestic savings rate of country *i* at time *t*,
- O_{i,t} denotes the log ratio of imports plus exports to GDP of country i at time t,

and the disturbance term $\varepsilon_{i,t}$ is assumed to be independent Gaussian with $E(\varepsilon_{i,t}) = 0$, and covariance $E(\varepsilon_{i,t}\varepsilon'_{i,t}) = \Omega_i$. We assume that the intercept c captures the effects of the technology parameters. The assumption that c is identical across countries is rather restrictive. Unfortunately, at the moment there is no technique available to determine an endogenous threshold in a real panel data context with individual effects. Because we are also interested in the contemporaneous effects (assuming them to be the same for each individual country), we rewrite (3) in the following structural form

$$X_{i,t} = \mu + B_0 X_{i,t} + B_1 X_{i,t-1} + u_{i,t}$$
(4)

where $u_{i,t}$ is again independent Gaussian with zero mean but with diagonal covariance matrix $E(u_{i,t}u'_{i,t}) = diag(\sigma^2_{i,1}, \sigma^2_{i,2}, \sigma^2_{i,3}) \equiv \Lambda_i$, where $\Omega_i = \Gamma \Lambda_i \Gamma'$ and Γ is lower triangular with ones on the main diagonal. We then have $\mu = \Gamma^{-1}c$, $B_0 = (I - \Gamma^{-1})$ and $B_1 = \Gamma^{-1}A_1$. Observe that B_0 is lower triangular with zeros on the main diagonal⁵.

To test the hypothesis that the relation between openness and GDP growth changes if the savings level exceeds a certain level we contrast the benchmark model in (4) with the following threshold VAR model:

$$X_{i,t} = (\mu + B_0 X_{i,t} + B_1 X_{i,t-1}) I(S_{i,t-2} \le \gamma)$$
(5)

$$+(\bar{\mu}+B_0 X_{i,t}+B_1 X_{i,t-1})I(S_{i,t-2}>\gamma)+w_{i,t}$$

where $I(S_{i,t-2} > \gamma)$ is an indicator function which equals one when the inequality holds and zero otherwise. The unknown threshold parameter γ indicates at which level of savings the change takes place. The use of $S_{i,t-2}$ instead of $S_{i,t}$ in the indicator function allows us to consider the savings level pre-determined. We proceed by introducing tests for the existence of an endogenous threshold as well as recently developed techniques for estimation of (5).

⁵For further details on the problem of identification, see Lütkepohl [28].

3 Econometric considerations

The economectric analysis of the above models requires a number of nonstandard techniques which are discussed in detail in this section. In particular, we discuss the following topics: tests for the existence of a threshold, estimation and inference of the threshold VAR model, and nonlinear impulse response analysis.

The econometric analysis of the benchmark VAR model (4) is essentially the analysis of a vector autoregressive model with pooled coefficients. This analysis is similar to that of a large VAR with restrictions on the coefficients of the lag polynomial. Since we assume T to be large we can apply the standard asymptotic theory on stationary vector autoregressions taking into account the pooling restrictions⁶.

3.1 Testing for the existence of a threshold

To test for the existence of a threshold we choose the following setup. Let the observable variables be denoted by (Y_i, X_i) for i = 1, ..., N. X_i is a $(1 \times k_x)$ vector and Y_i is a scalar variable. Let the following relation hold:

$$Y_i = X_i \beta_1 + Z_i I(q_i > \gamma) \beta_2 + \varepsilon_i \tag{6}$$

where Z_i is a $(1 \times k_z)$ vector, q_i is a one dimensional variable and γ is a scalar which we assume to be contained in a compact subset Γ of R. We assume that the variables in Z_i are also contained in X_i , i.e., Z_i is a subvector of X_i and thus observable as well. Finally, we assume ε_i to be a zero mean i.i.d distributed random variable with finite variance σ^2 . As noted above, I(.)denotes an indicator function which determines a possible break between observations satisfying the inequality condition and those not satisfying the condition.

There are several problems which complicate the analysis. First, under the null hypothesis of no threshold (i.e., $\mu = \bar{\mu}$, $B_0 = \bar{B}_0$, and $B_1 = \bar{B}_1$), the threshold parameter γ is not identified, and we therefore cannot apply standard hypothesis testing theory. To solve this issue, we derive a new test statistic appropriate for our case. This new test statistic builds upon and is related to versions of the Wald, LM, and LR statistics introduced by Andrews, Lee and Ploberger [1], Andrews and Ploberger [2] and Hansen [21], [22]. Second, most of the existing test statistics apply to time series and/or cross section data. The test statistic developed in this paper is well suited

⁶See Lütkepohl [28] or [20] for an introduction to this analysis.

for a panel data set. Third, some of the existing procedures are tedious to calculate due to bootstrapping methods, while others involve the choice of some rather subjective parameters. In contrast, our simple test statistic avoids both problems.

Our test procedure can be derived in two ways, a Wald and an LM version. The LM version of the test is presented below, while the Wald version is derived in Appendix B. To derive the LM based test statistic, we consider estimation of (5) under H_0 in which case the restricted OLS estimator is given by

$$\hat{\beta}_{1}^{r} = (X'X)^{-1}X'Y, \tag{7}$$

where $X = (X'_1, ..., X'_N)'$ and $Y = (Y'_1, ..., Y'_N)'$. Denoting the residuals of the restricted estimator by $u^r = Y - X \beta_1^r$ and introducing both the functions $Z_{(\gamma)} = (Z'_1 I(q_1 > \gamma), ..., Z'_N I(q_N > \gamma))'$ and the matrix $M = (I - X(X'X)^{-1}X')$ we observe that

$$Z'_{(\gamma)}u^{r} = Z'_{(\gamma)}(Y - X \stackrel{o}{\beta_{1}^{r}}) = Z'_{(\gamma)}(I - X(X'X)^{-1}X')Y$$
(8)

$$= Z_{(\gamma)}M(X\beta_1 + Z_{(\gamma)}\beta_2 + \varepsilon)$$
(9)

$$= Z'_{(\gamma)}M(Z_{(\gamma)}\beta_2 + \varepsilon).$$
(10)

This leads to

$$Z'_{(\gamma)}u^r = Z'_{(\gamma)}M\varepsilon \qquad under \quad H_0 \qquad (11)$$

$$Z'_{(\gamma)}u^r = Z'_{(\gamma)}MZ_{(\gamma)}\beta_2 + Z'_{(\gamma)}M\varepsilon \quad under \quad H_1$$
(12)

In Appendix B we show that the normalized stochastic function

$$z_N(\gamma) = \frac{1}{\sqrt{N}} Z'_{(\gamma)} u^r \tag{13}$$

converges, as $N \to \infty$, to a k_z dimensional Gaussian process $z(\gamma)$ with covariance kernel $\sigma(\gamma_1, \gamma_2)$ where

$$\sigma(\gamma_1, \gamma_2) = \lim_{N \to \infty} \frac{1}{N} \sigma^2 Z'_{(\gamma)} M Z_{(\gamma)}.$$

Since it is difficult to test whether the stochastic function (13) satisfies the behavior associated with the null hypothesis we use a transformation to summarize its behavior in one test statistic. We introduce the integrals

$$\int_{\Gamma} \mathcal{S}\left(z_N(\gamma) z'_N(\gamma)\right) d\gamma, \quad and \quad \int_{\Gamma} \mathcal{S}\left(\sigma_N(\gamma, \gamma)\right) d\gamma \tag{14}$$

where $S(\cdot)$ denotes the operator defined as the sum of all elements of its argument and $\sigma_N(\gamma, \gamma) = \frac{1}{N} \stackrel{\wedge}{\sigma}^2 Z'_{(\gamma)} M Z_{(\gamma)}$, where $\stackrel{\wedge}{\sigma}^2$ is a consistent estimate of σ^2 . The new test statistic, denoted by *BPH*, can now be defined as follows:

$$BPH = \int_{\Gamma} \mathcal{S}\left(z_N(\gamma) z'_N(\gamma)\right) d\gamma / \int_{\Gamma} \mathcal{S}\left(\sigma_N(\gamma, \gamma)\right) d\gamma.$$
(15)

Under the assumptions given in Appendix B, which are essentially the same assumptions as in Hansen [22], we can derive an asymptotic distribution of *BPH*. Since this asymptotic distribution depends on the data through the covariance kernel $\sigma(\gamma_1, \gamma_2)$ we follow Bierens and Ploberger [9] by using an upper bound to the true asymptotic distribution which is data independent⁷. The following critical regions are taken from Bierens and Ploberger:

$$P(W > 3.23) = 0.10, \quad P(W > 4.26) = 0.05, \quad P(W > 6.81) = 0.01.$$

The main advantage of the BPH test in comparison to the existing tests is its simplicity. In contrast to Hansen's F-test [22] no bootstrapping is needed. Instead we can use the data independent upper bound given in Bierens and Ploberger. Compared to the supremum, average, and exponential LM tests (supLM, aveLM and expLM, respectively) proposed by Andrews [3] and Andrews and Ploberger [2], our test statistic is independent of any nuisance parameters such as the number of regressors or the cut off levels of the threshold region. A disadvantage of our approach is that it can be conservative with the degree of conservatism depending on the problem at hand. Furthermore, it turns out that our test is less powerful than the related LM test statistics mentioned above⁸.

3.2 Estimation and inference on the threshold

If the above test statistics lead us to conclude that a threshold exists, we continue by estimating the threshold, γ , as well as the other coefficients of the model, namely β_1 and β_2 . The estimators for γ , β_1 and β_2 are the solution to the following nonlinear least squares problem:

$$(\widehat{\gamma}, \widehat{\beta}_1, \widehat{\beta}_2) = \arg \min_{\gamma \in \Gamma, \beta_1, \beta_2} (Y - X\beta_1 - Z_{(\gamma)}\beta_2)' (Y - X\beta_1 - Z_{(\gamma)}\beta_2).$$

 $^{^7\}mathrm{A}$ much more detailed discussion, in particular on the derivation of the upper bound, is given in Appendix B.

⁸Appendix B contains more detailed information about the power of the test as well as a result concerning a family of local alternatives. We also discuss the connection between our and the other LM test statistics.

Finding the global minimum can be achieved in two steps. First, we minimize the sum of squared errors for a fixed γ . Applying OLS gives us an estimate of the variance of the residuals, $\overset{\wedge}{\sigma}^2(\gamma)$. Second, we minimize $\overset{\wedge}{\sigma}^2(\gamma)$ over all $\gamma \in \Gamma$. The final estimates are then the OLS coefficients corresponding to the γ which minimizes $\overset{\wedge}{\sigma}^2(\gamma)$. Note that when the ϵ_i are i.i.d $N(0, \sigma^2)$, this estimator is also the MLE.

As is known from the literature (see, for example, Bai [4], Picard [33], and Chan [12]), the estimator for γ has a convergence rate of order n, which is much faster than the order of convergence (\sqrt{n}) for the other parameters of the model. The derivation of the asymptotic distribution of the estimator for γ is rather difficult, in particular when the change between β_1 and β_2 is considered to be fixed or relatively large. In this case the distance between the two parameters appears in the asymptotic distribution for $\hat{\gamma}$ which makes inference results almost impossible. However, under the assumption of a local alternative, i.e. a small difference between β_1 and β_2 , Hansen [22] is able to derive the asymptotic distribution of the Likelihood Ratio test statistic for $\gamma = \gamma_0$. He then uses this result to construct a confidence interval for γ . The confidence intervals presented in Table 1 and 4 below are based on his procedure.

3.3 Nonlinear impulse responses

Computing impulse response functions for nonlinear dynamic models is more complicated than computing impulse response functions for linear dynamic models for several reasons. One of the complications arises from the fact that in most cases there are no analytical results in the nonlinear case. This means that the impulse responses must be obtained numerically or need to be simulated. Further, it is much harder to present and investigate all the information contained in the impulse response of a nonlinear system. This is due to the fact that the response of a nonlinear system to a shock at time t_0 is path-dependent. The response depends in a nonlinear way on the history of the system, i.e., on the observations before the shock enters the system, and on the disturbances which enter the system between time t_0 and $t_0 + k$. Finally, the proportionality of a response to the size of the shock in a linear system does not hold in a nonlinear system⁹.

For the linear VAR we present the traditional impulse responses, including 2 standard-errors confidence bounds, based on 500 drawings from the

⁹More details on the problems of analyzing nonlinear impulse response can be found, amongst others, in Gallant, Rossi and Tauchen [17] and Potter [34].

distribution of the estimates of the parameters.

For the nonlinear threshold VAR we proceed as follows. We estimate an impulse response function corresponding to the Generalized Impulse Response function proposed by Potter [34] and also analyzed by Balke and Chang [7]:

$$GI(t_0, k, \delta, \Omega_{t_0-1}) = E(X_{t_0+k} | \Omega_{t_0-1}, \varepsilon_{t_0} = \delta) - E(X_{t_0+k} | \Omega_{t_0-1}),$$

where Ω_{t_0-1} is the history at time t_0 , and δ is the shock given to the system at time t_0 . To obtain this impulse response function we need to integrate out the future shocks $\varepsilon_{t_0+1}, ..., \varepsilon_{t_0+k}$, that is

$$GI(t_0, k, \delta, \Omega_{t_0-1}) = E_{\varepsilon_{t_0+1}, \dots, \varepsilon_{t_0+k}} \left(E\left(X_{t_0+k} | \Omega_{t_0-1}, \varepsilon_{t_0} = \delta, \varepsilon_{t_0+1}, \dots, \varepsilon_{t_0+k} \right) - E\left(X_{t_0+k} | \Omega_{t_0-1}, \varepsilon_{t_0+1}, \dots, \varepsilon_{t_0+k} \right) \right).$$

To do this we generate a large number of future zero mean i.i.d. normal shocks $\varepsilon_{t_0+1}^i, \dots, \varepsilon_{t_0+k}^i, i = 1, \dots, R$, and replace $GI(t_0, k, \delta, \Omega_{t_0-1})$ by

$$\widetilde{GI}(t_0, k, \delta, \Omega_{t_0-1}) = \frac{1}{R} \sum_{i=1}^{R} E(X_{t_0+k} | \Omega_{t_0-1}, \varepsilon_{t_0} = \delta, \varepsilon_{t_0+1}^i, ..., \varepsilon_{t_0+k}^i) \\ - \frac{1}{R} \sum_{i=1}^{R} E(X_{t_0+k} | \Omega_{t_0-1}, \varepsilon_{t_0+1}^i, ..., \varepsilon_{t_0+k}^i),$$

i.e., we average the impulse response function over the future shocks.

Next we notice that GI $(t_0, k, \delta, \Omega_{t_0-1})$ depends on Ω_{t_0-1} . In a threshold model it makes a big difference whether the system is close to the threshold level at t_0 , the time of the shock, or not. Therefore, we consider \widetilde{GI} $(t_0, k, \delta, \Omega_{t_0-1})$ conditional on different histories Ω_{t_0-1} . In our case we are especially interested in the behavior of the system for the high and low saving regimes. Therefore, we calculate \widetilde{GI} $(t_0, k, \delta, \Omega_{t_0-1})$ for the following three situations: unconditional on the regime we are at time t_0 , conditional on being in the high saving regime at time t_0 , and conditional on being in the low saving regime at time t_0 .

For each of the three situations we generate 100 histories, simulate the model, and leave out the first 500 observations to avoid initial observation problems. For each of the individual histories we calculate the generalized impulse response function based on 100 sets of future shocks (R = 100). Finally, we also take into account the uncertainty of the parameter estimates of the system. Therefore, we draw 50 sets of parameters from the asymptotic

distribution of the estimated parameters. For each set of parameters we replicate the above procedure. Therefore, each impulse response function presented in this paper is based on 5000 (= 50*100) simulations.

Since the impulse response function of a threshold model can be asymmetric, we investigate both negative and positive unit shocks. The results of our impulse response analysis are presented in Figures 2 to 10 in Appendix A. We present two sets of figures for each of the three classes of histories: we first present the average and the 95% most centered realizations of the impulse response function followed by the average impulse response function together with its two 95% confidence bounds.

4 The Data

Per capita GDP, savings and openness are taken from World Bank data [40]. GDP growth is the log difference of real per capita GDP in constant 1987 value of the local currency. The saving rate is the ratio of nominal gross domestic savings to nominal GDP (both in local currency), while openness to trade is the ratio of nominal exports plus imports to nominal GDP (both again in local currency). These data are available for all OECD countries as well as a number of developing countries. Excluding oil exporting countries, non-market economies, as well as countries with a population of less than one million, 58 countries remain in the sample¹⁰. The sample period ranges from 1960 to 1987¹¹.

We use log transforms of both GDP and openness to trade, a procedure that cannot be applied to the saving rate because the rate is negative in some years for some countries. Fortunately, scale problems do not arise since all variables are similarly scaled. Figure 1 in Appendix A displays the empirical distributions of the data. Clearly, the saving rate displays a good deal of variability which is necessary to verify our hypothesis of a changing relation between output growth and openness. Without such variability, testing and

¹⁰The countries in our sample are: Algeria, Argentina, Australia, Australia, Bangladesh, Belgium, Bolivia, Brazil, Canada, Chile, Colombia, Costa Rica, Cote d'Ivoire, Denmark, Egypt, El Salvador, Finland, France, Germany, Ghana, Greece, Guatemala, India, Indonesia, Ireland, Israel, Italy, Jamaica, Japan, Kenya, South Korea, Madagascar, Malaysia, Mauritius, Mexico, Morocco, Netherlands, New Zealand, Nigeria, Norway, Pakistan, Panama, Paraguay, Peru, Philippines, Portugal, Senegal, Singapore, Spain, Sri Lanka, Sweden, Switzerland, Thailand, Turkey, United Kingdom, United States, Uruguay, and Venezuela.

¹¹The sample period was chosen to allow for a direct comparison between our and earlier results such as Harrison [24] and Levine and Renelt [29].

estimation of a threshold level of saving would be futile.

5 Empirical Results

5.1 Long-run effects

We first consider the long-run relation between the saving rate and GDP growth. Using time-averaged variables for each country, i.e., $\Delta Y_i = \frac{1}{T} \sum_t \Delta Y_{i,t}$, $S_{i,-1} = \frac{1}{T} \sum_t S_{i,t-1}$ where T denotes the time dimension of the sample, simple OLS yields the following result:

$$\Delta Y_i = \begin{array}{ccc} 0.007 & + & 0.083 \ S_{i,-1} & R^2 = 0.13, \quad N = 58 \\ (0.006) & (0.029) \end{array}$$
(16)

The estimated coefficient for savings is positive and significant at the 5% level. Note that this result is based on the assumption of equality across countries. White's test for heteroscedasticity [39] yields a value of $1.972 (= N\tilde{R}^2)$ which is less than the critical value at the 5% level of the chi-square distribution with 3 degrees of freedom. We therefore cannot reject homogeneity. Further tests for normality of the residuals lead us to conclude that we cannot reject normality either.

Adding openness to trade to the above estimation model yields the following result:

$$\Delta Y_i = \begin{array}{ccc} 0.012 & + & 0.083 \ S_{i,-1} & + & 0.005 O_{i,-1} \ R^2 = 0.16, \quad N = 58\\ (0.007) & (0.029) & (0.003) \end{array}$$
(17)

Clearly, the effect of openness is small and insignificant! Again, we cannot reject normality nor homogeneity of the residuals (White's test yields $N\tilde{R}^2 = 2.826$).

Next we test for the existance of a threshold. In addition to our own test given in (15) and denoted by BPH, we calculate three LM tests (aveLM, expLM, supLM) as well as Hansen's F-test [22] denoted by HansenF. Since the regressors are averages over time, the threshold at a certain time would be conditioned on future values of the regressors producing inconsistent estimates. To avoid this problem we use the saving rate of 1960 as the threshold variable. Table 1 contains the values for the five test statistics, their corresponding p-values, as well the threshold estimate and its 95% confindence interval.

	Test Statistic	p-value
BPH	5.44	0.01
expLM	3.92	0.05
aveLM	7.05	0.01
supLM	10.23	> 0.10
HansenF	13.98	0.074
	Estimate of threshold	95% conf. interval
γ	0.1442	(0.049, 0.248)

Table 1: Threshold Tests and Estimation

As explained above, to conduct threshold tests based on the three LM tests as well as Hansen's F-test, we need to cut off a certain fraction, π , both at the top and the bottom of the empirical distribution of the saving rate. Based on 1960 data for the saving rate, this amounts to a search on the interval [0.095, 0.276] for $\pi = 15\%$. With the exception of the supLM test, all tests reject the null of no threshold at the 10% significance level. The resulting estimate for the threshold is 0.144. We use this value to split the sample and reestimate the above equation for each subsample. For the low savings countries ($\gamma \leq 0.144$) we find

$$\Delta Y_i = \begin{array}{ccc} 0.001 & + & 0.186 \ S_{i,-1} & + & 0.005 \\ (0.011) & (0.058) & (0.005) \end{array} \\ \end{array}$$

(18)

(19)

and for the high savings countries ($\gamma > 0.144$) we obtain

$$\Delta Y_i = -0.003 + 0.107 S_{i,-1} - 0.005 O_{i,-1} R^2 = 0.22, \quad N = 36$$

$$(0.011) \quad (0.040) \quad (0.004)$$

with a joined $R^2 = 0.32$. Interestingly, the sign of the coefficient for openness varies between the two subgroups of countries, but both signs are insignificant. Also, compared to the benchmark model, the estimated coefficient for the average saving rate is larger for both subgroups.

5.2 VAR analysis: estimation, identification and testing

First we investigate the stationarity for each of the variables. For this we use the test statistics developed by Im et al. [25]. They develop unit root tests for heterogeneous panels specified by

$$x_{i,t} = (1 - \phi_i)\mu_i + \phi_i x_{i,t-1} + \varepsilon_{i,t}$$

where $E(\varepsilon_{i,t}) = \sigma_i^2$. They test the hypothesis $\phi_i = 1$ for all *i* against $\phi_i < 1$ for all *i*. Since we impose that $\phi_i = \phi$ and $\mu_i = \mu$ for all *i* in our model, we can use this procedure to test for unit roots. Furthermore, since their test is not based on these restrictions it makes our results robust against this type of misspecification. The results, given in Table 2, indicate that we can formally reject the hypothesis of a unit root in GDP growth and the saving rate. For openness, however, we cannot reject the hypothesis of a unit root. This result has some implications for the model proposed in (4) and (5).

Table 2: Tests for unit roots.

Variable	$LR_{26}(0,0)$	p-value
GDP growth	2.85	0.002
saving rate	3.164	< 0.001
openness	-0.884	0.199

Since $O_{i,t}$ is I(1) and $\Delta y_{i,t}$ and $S_{i,t}$ are I(0) we have to put in place restrictions on the coefficients of openness to obtain a stationary system. Therefore we restrict the coefficients of current and lagged openness to be equal but of opposite sign in all equations including the threshold model given in (5). Imposing this restriction allows us to restate (4) and (5) in terms of $\Delta O_{i,t} = O_{i,t} - O_{i,t-1}$:

$$X_{i,t}^* = \mu + B_0 X_{i,t}^* + B_1 X_{i,t-1}^* + u_{it}$$
⁽²⁰⁾

and

$$X_{i,t}^* = (\mu + B_0 X_{i,t}^* + B_1 X_{i,t-1}^*) I(S_{i,t-2} \le \gamma)$$
(21)

 $+(\bar{\mu}+\bar{B}_0 X_{i,t}^*+\bar{B}_1 X_{i,t-1}^*)I(S_{i,t-2}>\gamma)+w_{it}$

where $X_{i,t}^* = (\Delta O_{it}, S_{i,t}, \Delta Y_{it})'$ and where the following parameter restrictions have been imposed:

$$B_0, B_0, \bar{B}_0: \begin{bmatrix} 0 & 0 & 0 \\ . & 0 & 0 \\ . & . & 0 \end{bmatrix}, \text{ and } B_1, B_1, \bar{B}_1: \begin{bmatrix} 0 & . & . \\ 0 & . & . \\ 0 & . & . \end{bmatrix}.$$

	Dej	pendent Vari	iable:
regressors	$\Delta y_{i,t}$	$S_{i,t}$	$\Delta O_{i,t}$
0	0.005*	0.011*	0.008
C	(0.002)	(0.002)	(0.007)
S	0.522*		
$D_{i,t}$	(0.031)		
10	0.033*	0.014*	
$\Delta O_{i,t}$	(0.007)	(0.005)	
An	0.302*	0.066*	0.181*
$\Delta g_{i,t-1}$	(0.022)	(0.016)	(0.062)
$S_{i,t-1}$	-0.466^{*}	0.942*	0.011
	(0.031)	(0.007)	(0.032)
	$R^2 = 0.41$	$R^2 = 0.98$	$R^2 = 0.009$

Table 3: Estimation results: linear VAR.

standard errors in parentheses

*: significant at 5% level

We present the estimation results of (20) in Table 3. As explained above we can estimate this system as three separate equations with GDP growth, the saving rate, and change in openness as the respective dependent variables (standard errors are given in parentheses).

In the first regression (with GDP growth as the dependent variable) all the explanatory variables are significant at the 5% level. As the next column in Table 3 reveals (with the savings rate as the dependent variable) a change in openness has a significant impact on the savings rate. Therefore, a change in openess has also an indirect impact on GDP growth through its effect on savings. In the third regression (with the change of openness as dependent variable) lagged GDP growth is positive and significant, while the saving rate has no significant impact on the change in openness. The complexity of the relation between openness and growth will become clearer when we present the results of the impulse response analysis in the next section.

To check the validity of this model we perform the Roy-Zellner test for poolability. For the GDP growth and the saving rate regression we obtain F-values of 3.134 (p-value <0.001) and 2.426 (p-value <0.001), respectively, which means that we reject the null hypothesis that all coefficients are the same for each country. This implies that the linear model is misspecified, which is exactly what we would expect if a threshold exists. For the change in openness we do not reject the pooling hypothesis (p-value is equal to 0.4717),

a result which may not be surprising given the low R^2 of this regression.

Next, we test for the existence of a threshold using the same tests as presented in Table 1. In a VAR system as the one we analyze there are two ways to model the breakpoint. A threshold can be modeled separately for each of the three equations. Not surprisingly, we may find a different threshold for each equation in this case. Alternatively, we can restrict the threshold to be the same for all three equations. We pursue both approaches presented in Table 4.

	Threshold Test Statistics			
	Dependent Variable:			Restricted VAR
	$\Delta y_{i,t}$	$S_{i,t}$	$\Delta O_{i,t}$	
BPH	1.62	2.40	3.23*	1.85
expLM	17.80**	10.74**	7.41*	34.50**
aveLM	15.78**	12.29**	11.25**	40.00**
supLM	41.92**	27.66*	19.51*	77.66**
HansenF	45.15**	30.73**	21.02*	49.82**
	Threshold Estimation			
γ	0.181	0.240	0.154	0.192
95% conf. inter.	(0.177, 0.204)	(0.229, 0.249)	(0.075, 0.227)	(0.180, 0.247)

Table 4: Threshold tests and estimation in VAR.

*: significant at 5% level

**: significant at 1% level

The aveLM, expLM and supLM tests are based on $\pi = 0.05$. The F-test of Hansen is derived under the assumption of maintained homogeneity of the variances of the disturbances among the two groups. First, notice that, as expected (see appendix B), the BPH test suffers from a lack of power as compared to the other tests. Second, the threshold estimates differ across the three equations as expected. In the absence of a reasonable economic interpretation for the different threshold values, we restrict our analysis to the model where the thresholds are restricted to be the same across equations.

Based on the threshold level of 0.192, we estimate the resulting threshold model. The results are presented in Table 5 (standard errors are in parentheses). There are 670 observations in the low savings regime and 838 observations in the high savings regime. Furthermore, while 9 countries are in the low saving regime at every point in time, 15 countries are always in the high saving regime. All other countries switch regimes at some point(s) during the sample period.

	$S_{i,t-2} \ll 0.19, N = 670$		$S_{i,t-2} > 0.19, N = 838$		= 838	
	Dependent Variable:					
regressors	$\Delta y_{i,t}$	$S_{i,t}$	$\Delta O_{i,t}$	$\Delta y_{i,t}$	Si,t	$\Delta O_{i,t}$
0	0.005	0.011*	-0.019	0.000	0.013*	-0.016
C	(0.004)	(0.003)	(0.012)	(0.004)	(0.003)	(0.014)
$S_{i,t}$	0.333*			0.704*		
	(0.041)			(0.046)		
10	0.0027	-0.006		0.033*	0.031*	
$\Delta O_{i,t}$	(0.009)	(0.008)		(0.011)	(0.007)	
Aa	0.368*	0.050	-0.073	0.253*	0.072*	0.301*
$\Delta g_{i,t-1}$	(0.035)	(0.027)	(0.103)	(0.030)	(0.020)	(0.104)
$S_{i,t-1}$	-0.278^{*}	0.937*	0.242*	-0.622^{*}	0.931*	-0.061
	(0.045)	(0.018)	(0.082)	(0.047)	(0.014)	(0.092)
	$R^2 = 0.33$	$R^2 = 0.93$	$R^2 = 0.02$	$R^2 = 0.44$	$R^2 = 0.98$	$R^2 = 0.024$

Table 5: Estimation results: threshold model.

standard errors in parentheses

*: significant at 5% level

The most important result is that a change in openness has a positive and significant impact on saving in the high saving regime, while it is negative though not significant in the low saving regime. In addition, both current and lagged savings have larger impact on GDP growth in the high saving than in the low saving regime. Finally, lagged GDP growth has a positive and significant impact on a change in openness in the high saving regime, while its impact is negative though not significant in the low saving regime. In contrast, lagged savings is positive and significant in the low and negative and insignificant in the high saving regime.

5.3 VAR analysis: Impulse responses

The impulse response analysis is based on the coefficient estimates presented in Table 3 for the linear model and in Table 5 for the nonlinear threshold model.

Observe that the threshold tests presented in the previous section indicate the existence of a threshold even if only one of the coefficients of μ , B_0 or B_1 changes significantly between the two regimes. Clearly, in this case one could obtain efficiency gains by restricting all other coefficients to be the same. Since we do not restrict the coefficients in this way, a potential efficiency loss is possible, not only for the coefficient estimates but for the impulse response functions as well. However, any potential efficiency loss does not affect the validity of our analysis or the possible outcomes.

We start by analyzing the linear impulse responses to unit shocks. It is worth noting that in our model the innovations are already orthogonal. The impulse responses for the linear VAR from Table 3 are given in Figure 2. Observe that there are significant responses of GDP growth and savings to a change in openness. The response in GDP growth, however, is small and during a short time even negative. As expected, a shock in savings produces a larger reaction of output growth than a shock in openness. This confirms the theoretical notion that the saving level is a more important determinant of output growth than openness.

We now analyze the nonlinear impulse responses to a unit shock. Figures 3 to 5 present the average nonlinear impulse responses as well as the 95% most centered realizations to a unit shock in openness depending on the starting level of savings. Figure 3 gives the results unconditional whether the starting level of savings is above or below the threshold, Figure 4 conditional on being above, and Figure 5 conditional on being below the threshold.

In each figure, the first row gives the responses to a positive unit shock, while the second row traces the effects of a negative unit shock. The average impulse response with 2 standard deviation confidence bounds, based on the same simulation, are given in the Figures 6 to 8.

Comparing the nonlinear impulse response functions (Figure 6, first row) with those from the linear model (Figure 2, first column), we notice that the responses of the saving rate and GDP growth to a change in openness are substantially different. In the linear case the effect of the shock quickly disappears, while in the nonlinear case the effect is still significant after 24 periods. Even more important, the effect of a shock of openness on GDP growth becomes negative after a few periods in the linear case but remains positive throughout in the nonlinear case.

Next we analyze the three situations of the nonlinear model. As expected, the magnitude of the unconditional response is between the two conditional responses. Further, we observe clear differences between responses originating from a high saving regime and responses originating from the low saving regime. In the high saving regime there is always a positive effect on savings resulting from a positive shock in openness. This is in contrast to the response initiated from a low saving regime where a positive shock initially has a significant negative impact on the saving rate, while a negative shock has a positive impact on the saving rate. Clearly, this striking asymmetry of responses to positive and negative shocks in the low saving regime cannot be generated by a linear system. Finally, notice that the response in the high saving regime is larger than the response in the low savings regime. In particular, the response of GDP growth to a positive shock in openness is larger in the high saving regime. Even after 10 years, it is still positive, while in the low saving regime the initially positive response becomes insignificant after 4 years.

To test whether we can observe similar effects for specific countries we calculate the impulse responses for the United States, a notorious low savings country, and Japan, a country known for its high saving rate. The impulse responses are based on the parameter estimates and asymptotic distributions from the model given in Table 5 together with the respective histories of the two countries. For each country we use the last years in the sample, 1985-1987, as the time of the shock. We simulate 100 sets of parameters and then expose the system to a unit shock in openness. The results are shown in Figures 9 and 10.

The figures reveal a number of interesting differences. First, a positive shock in openness has a positive and significant impact on Japan's saving rate, even after 24 periods. In comparison, the same shock has essentially no significant impact on the saving rate in the United States. In addition, a positive shock in openness has a positive impact on GDP growth in Japan for all periods, while the same shock has a much smaller, largely insignificant impact on U.S. growth.

6 Summary and Conclusion

In this paper we test a simple hypothesis: does a country's saving rate matter both over time and across countries for the interaction between growth and openness? Using a dynamic model for GDP growth, the saving rate and change in openness to trade, we find that it indeed matters. Countries with high saving rates experience a growth-enhancing effect from an increase in openness, while countries with low saving rates do not experience such an effect. The differences between low and high saving countries are evident from the estimation of the threshold VAR models. In the high saving regime, openness has a positive effect on growth, while the same effect in the low saving regime is negative though insignificant. There are striking differences in the impulse response functions between the two regimes as well. In the high saving regime, a positive shock in openness leads to higher GDP growth in all subsequent years, while in the low saving regime the effect on GDP growth is substantially smaller in size and becomes insignificant after a few periods.

Our empirical findings have nontrivial policy implications. Clearly, low saving rates are a double curse for a country. On the one hand, a low national saving rate directly diminishes the domestic growth fundamentals of the economy. Furthermore, as our analysis indicates, it also undermines the potential growth effects of increased openness to world trade experienced by high saving countries. Even worse, for some countries a higher volume of trade may even reduce the GDP growth rate which is already relatively low due to the low national savings rate.

In addition to the empirical results, the paper also adds to the theoretical econometric literature by introducing a new simple test for endogenous threshold models. The test has the advantage that it can be calculated efficiently since no bootstrapping is needed. Another advantage is that the test does not involve the choice of any subjective parameters such as the number of regressors or the cut off levels of the threshold region.

References

- Andrews, D.W.K., Lee, I. and W. Ploberger, (1996), "Optimal change point tests for normal linear regression", Journal of Econometrics 70, 9-38.
- [2] Andrews, D.W.K. and W. Ploberger, (1994), "Optimal tests when a nuisance parameter is present only under the alternative", Econometrica 62, 1383-1414.
- [3] Andrews, D.W.K., (1993), "Tests for parameter instability and structural change with unknown change point", Econometrica, 61, 821-856.
- [4] Bai, J., (1995), "Estimation of structural change based on Wald-type statistics", Working Paper, Department of Economics, M.I.T.
- [5] Balassa, B., (1985), "Exports, policy choices, and economic growth in developing countries after the 1973 oil price shock," Journal of Development Economics 18(2), 23-35.
- [6] Balassa, B., (1989), "Outward orientation," in Handbook of Development Economics, edited by H. Chenery and T.N. Srinivasan, Amsterdam: North-Holland, 1645-89.
- [7] Balke, N.S., and C. Chang, (1995), "Credit and economic activity: shocks or propagation mechanism?", Working Paper, Southern Methodist University.
- [8] Baltagi, B. H., (1995), "Econometric analysis of panel data", John Wiley and Sons, Chichester
- [9] Bierens, H.J. and W. Ploberger, (1995), "Asymptotic theory of integrated conditional moment tests", Econometrica, forthcoming.
- [10] Bosworth, B., (1993), "Saving and investment in a global economy", Washington: Brookings Institution.
- [11] Carroll, C.D. and D.N. Weil, (1993), "Saving and growth: a reinterpretation", NBER Working Paper No. 4470.

- [12] Chan, K.S., (1993), "Consistency and limiting distribution of the least squares estimator of a threshold autoregressive model", Annals of Statistics 21, 4520-533.
- [13] Davidson, R., and J.G. MacKinnon, (1993), "Estimation and inference in econometrics", Oxford University Press, New York.
- [14] Edwards, S., (1992), "Trade orientation, distortions, and growth in developing countries," Journal of Development Economics 39, 31-57.
- [15] Edwards, S., (1993), "Openness, trade liberalization, and growth in developing countries," Journal of Economic Literature 31, 1358-1393
- [16] Feenstra, R. (1996), "Trade and uneven growth," Journal of Development Economics 49(1), 199-227.
- [17] Gallant, A., Rossi, P.E., Tauchen, G., (1993), "Nonlinear dynamic structures", Econometrica 61(4), 871–908.
- [18] Gordon, R.H., and A.L. Bovenberg, (1996), "Why is capital so immobile internationally? Possible explanations and implications for capital income taxation", American Economic Review 86(5), 1057-1075.
- [19] Gupta, K.L. and M.A. Islam, (1983), "Foreign capital, savings, and growth", International Studies in Economics and Econometrics, Vol. 9, D. Reidel Publishing Company, Dordrecht, Holland.
- [20] Hamilton, J.D., (1994), "Time series analysis", Princeton University Press; Princeton, NJ.
- [21] Hansen, B.E., (1995), "Endogenous sample splitting and the estimation of thresholds", manuscript, Boston College.
- [22] Hansen, B.E., (1996a), "Inference when a nuisance parameter is not identified under the null hypothesis", Econometrica 64(2), 413-430.
- [23] Hansen, B.E., (1996b), "Sample Splitting and Threshold Estimation", manuscript, Boston College.
- [24] Harrison, A., (1996), "Openness and growth: A time-series, crosscountry analysis for developing countries", Journal of Development Economics 48, 419-447.
- [25] Im, K., Pesaran, M. H., and Shin, Y., (1996), "Testing for unit roots in heterogeneous panels", manuscript, University of Cambridge.

- [26] Jorgenson, D. W. and M. S. Ho, (1993), "Trade policy and U.S. economic growth," HIER Discussion Paper No. 1634.
- [27] Lee, J.-L., (1995), "Capital goods imports and long-run growth", Journal of Development Economics 48, 91-110.
- [28] Lütkepohl, H., (1991), "Introduction to multiple time series analysis", Springer Verlag, Berlin-Heidelberg.
- [29] Levine, R. and D. Renelt, (1992), "A sensitivity analysis of cross-country growth regressions", American Economic Review 82(4), 942-963.
- [30] Maddison, A., (1992), "A long run perspective on saving", in Saving Behavior: Theory, International Evidence, and Policy Implications, edited by E. Koskela and J. Paunio, Cambridge: Blackwell, 27-59.
- [31] Osang, T. and A. Pereira, (1996), "Savings, volume of trade, and growth," Review of International Economics 5(3),
- [32] Osang, T. and A. Pereira, (1996), "Foreign growth and domestic performance in a small open economy", Journal of International Economics 43(3/4), 499-512.
- [33] Picard, D., (1985), "Testing and estimating change points in time series model", Stochastic Processes and Their Applications 19, 297-303.
- [34] Potter, S., (1995), "A non-linear approach to US GNP", Journal of Applied Econometrics 10, 109-125.
- [35] Quah, D. and J.E. Rauch, (1990), "Openness and the rate of economic growth", manuscript (MIT, Cambridge, and UCSD, San Diego.).
- [36] Rivera-Batiz, L. and P. Romer, (1991), "Economic integration and endogenous growth", Quarterly Journal of Economics 106(2), 531-555.
- [37] Thomas, V., N. Halevi and J. Stanton, (1991), "Does policy reform improve performance?" Background paper for World Development Report 1991.
- [38] Turnovsky, S.J. (1996), "Fiscal policy, growth, and macroeconomic performance in a small open economy", Journal of International Economics 40, 41-66.

- [39] White, H., (1980), "A heteroscedasticity-consistent covariance matrix estimator and a direct test for heteroscedasticity", Econometrica 48, 817-838.
- [40] World Data 1995, (1996), World Bank Indicators on CD-Rom, The World Bank, Washington, DC.

Appendix A



Figure 1: Distribution of the data; frequencies are given on the vertical axis



Figure 2: Impulse response functions to unit shocks with 2 standard deviation confidence bounds for the benchmark VAR model.



Figure 3: Impulse response functions (average and 95% most centered realizations) to a unit shock in openness for threshold VAR model: initial saving rates unconditional on saving regime



Figure 4: Impulse response functions (average and 95% most centered realizations) to a unit shock in openness for threshold VAR model: initial saving rates conditional on high saving regime



Figure 5: Impulse response functions (average and 95% most centered realizations) to a unit shock in openness for threshold VAR model: initial saving rates conditional on low saving regime



Figure 6: Average impulse response functions to a unit shock in openness with two standard deviation confidence bounds for threshold VAR model: initial saving rates unconditional on saving regime



Figure 7: Average impulse response functions to a unit shock in openness with two standard deviation confidence bounds for threshold VAR model: initial saving rates conditional on high saving regime



Figure 8: Average impulse response functions to a unit shock in openness with two standard deviation confidence bounds for threshold VAR model: initial saving rates conditional on low saving regime



Figure 9: Average impulse response functions to a unit shock in openness with two standard deviation confidence bounds for threshold VAR model: Japan



Figure 10: Average impulse response functions to a unit shock in openness with two standard deviation confidence bounds for threshold VAR model: United States

Appendix B

In section 3.1 we introduced a statistic related to the traditional LM test statistic. Here we derive a test statistic related to the traditional Wald statistic. The Wald statistic is based on the unrestricted parameter estimates of (β_1, β_2) ; recall that we test the restriction $\beta_2 = 0$ against the alternative $\beta_2 \neq 0$. For fixed scalar γ the OLS estimator for β_2 is given by

$$\hat{\beta}_2(\gamma) = \left(Z'_{(\gamma)}MZ_{(\gamma)}\right)^{-1}Z'_{(\gamma)}MY,\tag{22}$$

where M = (I - X(X'X)X') . We now define the normalized stochastic function as

$$z_N(\gamma) = \frac{1}{\sqrt{N}} \left(Z'_{(\gamma)} M Z_{(\gamma)} \right) \stackrel{\wedge}{\beta}_2(\gamma) = \frac{1}{\sqrt{N}} Z'_{(\gamma)} M Y.$$
(23)

Observe the replacement of u^r , the residuals based on the restricted estimator of β_1 in (13) by Y in (22). We now have

$$Z'_{(\gamma)}Y = Z'_{(\gamma)}M\varepsilon \qquad under \quad H_0, \qquad (24)$$

$$Z'_{(\gamma)}Y = Z'_{(\gamma)}MZ_{(\gamma)}\beta_2 + Z'_{(\gamma)}M\varepsilon \qquad under \quad H_1.$$
(25)

Comparing (24) and (25) to (11) and (12) in Section 3.1 reveals that the asymptotic behavior of the test static based on the unrestricted estimator is identical to the one based on the restricted estimator. Before we derive this asymptotic distribution we discuss some results concerning the power of the test.

The power of the BPH test: Consider the model

$$Y = X\beta_1 + Z_{(\gamma)}\beta_2 + u, \qquad u \backsim N(0, \sigma^2 I)$$

with β_1 a k_x dimensional parameter vector. For fixed γ the LM statistic for the null hypothesis ($\beta_2 = 0$) is given by

$$LM(\gamma) = \frac{(Y - X \beta_1)' Z_{(\gamma)} (Z'_{(\gamma)} M Z_{(\gamma)})^{-1} Z'_{(\gamma)} (Y - X \beta_1)}{Y' M Y}$$

= $\frac{u^{r'} Z_{(\gamma)} (Z'_{(\gamma)} M Z_{(\gamma)})^{-1} Z'_{(\gamma)} u^r}{\sigma^2}$

where $\widetilde{\beta_1}$ is the estimator based on the restricted estimate, while M and u^r are based on $\widetilde{\beta_1}$. Since $u^r = MY = Mu$ we find that under H_o

$$LM(\gamma) = \frac{u^{r'}MZ_{(\gamma)}(Z'_{(\gamma)}MZ_{(\gamma)})^{-1}Z'_{(\gamma)}Mu^{r}}{\overset{\wedge}{\sigma^{2}}}$$
(26)

$$= \frac{1}{\frac{\wedge}{\sigma^2}} trace((Z'_{(\gamma)}MZ_{(\gamma)})^{-1}Z'_{(\gamma)}Muu'MZ_{(\gamma)})$$

It is now straightforward to explain the relation between our test statistic and the exponential–, average–, and supremum LM statistics. All these test statistics are based on functionals of $LM(\gamma)$. In contrast our test makes use of the functional

$$\frac{1}{\overset{\wedge}{\sigma^2}}\mathcal{S}\left(z_N(\gamma)z'_N(\gamma)\right) = \frac{1}{\overset{\wedge}{\sigma^2}}\mathcal{S}\left(Z'_{(\gamma)}Muu'MZ_{(\gamma)}\right)$$
(27)

where $\hat{\sigma}^2$ is the same as in (26).

Under the null hypothesis the matrix $(Z'_{(\gamma)}MZ_{(\gamma)})^{-1}Z'_{(\gamma)}Muu'MZ_{(\gamma)}$ is distributed as a diagonal matrix with diagonal elements which are i.i.d $\sigma^2 \mathcal{X}^2$ distributed. This implies that the LM statistic $LM(\gamma)$ is asymptotically distributed as a standard normal chi-square statistic which is independent of the data. This means that critical values can be tabulated. Clearly $Z'_{(\gamma)}Muu'MZ_{(\gamma)}$ in (27) is asymptotically not diagonal but still depends on the data through $(Z'_{(\gamma)}MZ_{(\gamma)})^{-1}$. To take into account the off diagonal elements we introduce the $\mathcal{S}(.)$ operator.

Since $\frac{1}{\sigma^2} S(z_N(\gamma) z'_N(\gamma))$ depends on the data, $\int \frac{1}{\sigma^2} S(z_N(\gamma) z'_N(\gamma)) d\gamma$ does as well. It is shown below that a data independent upper bound of the asymptotic distribution of $\int \frac{1}{\sigma^2} S(z_N(\gamma) z'_N(\gamma)) d\gamma$ can be given. An additonal advantage is that the number of regressors does not affect the asymptotic distribution; this in contrast to the LM statistic which depends on k_x . Using an upper bound automatically implies that the test statistic will exhibit some conservatism. The degree of conservatism depends on the data.

Asymptotic distribution of the BPH statistic: Using the notation from Section 3.1 we again find

$$z_N(\gamma) = \frac{1}{\sqrt{N}} Z'_{(\gamma)} u^r = \frac{1}{\sqrt{N}} Z'_{(\gamma)} M \varepsilon \qquad under \quad H_0,$$

$$z_N(\gamma) = \frac{1}{\sqrt{N}} Z'_{(\gamma)} u^r = \frac{1}{\sqrt{N}} Z'_{(\gamma)} M Z_{(\gamma)} \beta_2 + \frac{1}{\sqrt{N}} Z'_{(\gamma)} M \varepsilon \quad under \quad H_1$$

Under the assumptions given below $\frac{1}{\sqrt{N}}Z'_{(\gamma)}M\varepsilon$ converges weakly to a zero mean k_z dimensional Gaussian process. This means that under H_0 the process $z_N(\gamma)$ is also a Gaussian process with zero mean. Under H_1 , however, there is an additional term $\frac{1}{\sqrt{N}}Z'_{(\gamma)}MZ_{(\gamma)}\beta_2$. Given the assumptions below it can be shown that, for fixed γ ,

$$\lim_{N \to \infty} \frac{1}{N} Z'_{(\gamma)} M Z_{(\gamma)} = \sigma^2 \Omega_{(\gamma)}.$$

Clearly, $\frac{1}{\sqrt{N}}Z'_{(\gamma)}MZ_{(\gamma)}\beta_2$ diverges to plus or minus infinity when $\beta_2 \neq 0$, a fact that makes our test consistent. In addition, we can consider our test under some local alternative. We may assume that the parameter β_2 is given by $\frac{\beta_2}{\sqrt{N}}$ with $\tilde{\beta}_2 \neq 0$, i.e. we assume β_2 to be very small initially and to become even smaller as the sample size increases. The first step to derive the asymptotic distribution of the test statistic and its behavior under the local alternatives requires the following theorem.

Theorem 1 Under H_1 , i.e. $\beta_2 \neq 0$, and given the assumptions stated below, we have $z_N(\gamma) \Rightarrow z(\gamma)$, where $z(\gamma)$ is a k_z dimensional Gaussian process with mean function

$$\eta(\gamma) = p \lim_{N \to \infty} \frac{1}{\sqrt{N}} Z'_{(\gamma)} M Z_{(\gamma)} \beta_2$$

and variance function

$$\sigma(\gamma_1, \gamma_2) = \lim_{N \to \infty} \sigma^2 \frac{1}{N} Z'_{(\gamma_1)} M Z_{(\gamma_2)}.$$

Further, by the continuous mapping theorem

$$iz_N \to iz = \int \mathcal{S}(z_N(\gamma)z'_N(\gamma)) d\gamma$$
 in distribution.

Proof. We postpone the proof until the end of the appendix.

We also need the following two lemmas which are versions of Mercer's Theorem $^{12}. \ \,$

Lemma 2 (Mercer 's Theorem) Let $\Psi(\gamma_1, \gamma_2)$ be a real valued positive semidefinite continuous function on $\Gamma \times \Gamma$, where Γ is a compact space, and let μ be a probability measure on Γ . The solutions λ_i and $\psi_i(\bullet)$, i = 1, 2, 3, ... of the Eigenvalue problem $\int \Psi(\gamma_1, \gamma_2)\psi_i(\gamma_2)d\mu(\gamma_2) = \lambda_i\psi_i(\gamma_1)$ are real valued and the function Γ has the series representation $\Psi(\gamma_1, \gamma_2) = \sum_{i=1}^{\infty} \lambda_i \psi_i(\gamma_1) \psi_i(\gamma_2)$, where the series involved converges uniformly on $\Gamma \times \Gamma$.

Lemma 3 Let the conditions of Lemma 2 be satisfied. The Eigenvalues λ_i are nonnegative and satisfy $\sum_{i=1}^{\infty} \lambda_i < \infty$. Moreover, the Eigenfunctions $\psi_i(\bullet)$ are continuous and can be chosen orthonormal and complete in the space $C(\Gamma)$ of continuous real functions on Ξ as well on the space $L_2(\mu)$ of squared

¹²The two lemmas are also stated in Bierens and Ploberger [9].

integrable functions with respect to μ , i.e. $\int \psi_i(\gamma)\psi_j(\gamma)d\mu(\gamma) = I(i = j)$, and every function ϕ in $C(\Gamma)$ or $L_2(\mu)$ can be written as

$$\phi(\gamma) = \sum_{i=1}^{\infty} g_i \psi_i(\gamma)$$
 a.s. $L_2(\mu)$,

with Fourier coefficients

$$g_i = \int \phi(\gamma) \psi_i(\gamma) d\mu(\gamma)$$

satisfying $\sum_{i=1}^{\infty} g_i^2 < \infty$.

We now apply these two lemmas to our statistic. Let Ψ in Lemma 2 be

$$\Psi(\gamma_1,\gamma_2)=\mathcal{S}\left(\sigma(\gamma_1,\gamma_2)
ight)$$
 .

Then the continuity of the Gaussian process $z(\gamma)$, and the compactness of Γ imply that $z(\gamma)$ is squared integrable. Further since the set $\{\psi_i, i = 1, 2, ..\}$ of Eigenfunctions is complete we can apply Parseval 's identity

$$\begin{split} \int \mathcal{S}\left(z(\gamma)z'(\gamma)\right)d\mu(\gamma) &= \int \mathcal{S}\left(z(\gamma)\right)^2 d\mu(\gamma) \\ &= \sum_{i=1}^{\infty} \left(\int \mathcal{S}\left(z(\gamma)\right)\psi_i(\gamma)d\mu(\gamma)\right)^2 \end{split}$$

Observe that $\phi(\gamma) = S(z(\gamma))$. By the fact that the sum of Gaussian processes is again Gaussian it follows that the Fourier coefficients

$$\int \mathcal{S}(z(\gamma))\psi_i(\gamma)d\mu(\gamma), \qquad i=1, 2, 3....$$
(28)

are Gaussian as well. For the characterization of their joint distribution we only need the covariances and means. The covariances are given by

$$E\left[\int \mathcal{S}\left(z(\gamma) - \eta(\gamma)\right)\psi_i(\gamma)d\mu(\gamma)\int \mathcal{S}\left(z(\gamma) - \eta(\gamma)\right)\psi_j(\gamma)d\mu(\gamma)\right]$$

=
$$\int \int \mathcal{S}\left(\sigma(\gamma_1, \gamma_2)\right)\psi_i(\gamma_1)\psi_j(\gamma_2)d\mu(\gamma_1)d\mu(\gamma_2) = \lambda_i I(i=j),$$

which means that the sequence (28) is independent. Further it is easy to see that the mean of the *i*-th element of the sequence (28) is just the *i*-th Fourier coefficient of $\eta(\gamma)$

$$\eta_i = \int \eta(\gamma) \psi_i(\gamma) d\mu(\gamma).$$

Note that the way in which we apply Lemma 2 and 3 is similar to Bierens and Ploberger [9]. From the above it directly follows that under the local alternative $\beta_2 = \frac{\tilde{\beta}_2}{\sqrt{N}}$ with $\tilde{\beta}_2 \neq 0$ the following theorem holds.

Theorem 4 Under H_1 , i.e $\beta_2 \neq 0$.we have

$$iz = \int \mathcal{S}(z(\gamma)z'(\gamma)) d\gamma \sim \sum_{j=1}^{\infty} (\eta_j + \epsilon_j \sqrt{\lambda_j})^2$$

where ϵ_i i.i.d N(0, 1) and λ_i and η_i are as described above.

Under the null we have

$$H_0: \quad T_0 = \sum_{j=1}^{\infty} \epsilon_j^2 \lambda_j, \tag{29}$$

where the λ_j depend on Ψ and are therefore data dependent. This implies that the asymptotic distribution of T_0 also depends on the data. However, using a theorem derived by Bierens and Ploberger we can obtain critical values which are data independent.

Theorem 5 (Theorem 5 of Bierens and Ploberger) Let ε_j be i.i.d N(0,1) and let

$$\bar{W} = \sup_{T \ge 1} \frac{1}{T} \sum_{j=1}^{T} \varepsilon_j^2.$$

For $\eta > 0, P(T_{H_0} > \eta E(T_{H_0})) \le P(\bar{W} > \eta)$, where T_{H_0} is the random variable defined by (29). Consequently, under the null hypothesis of $\beta_2 = 0$

$$\lim_{N\to\infty} P\left(iz_{H_0} > \eta \int \mathcal{S}(\sigma_N(\gamma,\gamma))d\gamma\right) \leq P\left(\bar{W} > \eta\right).$$

Bierens and Ploberger also simulate the distribution of \overline{W} . Using 10,000 replications they obtain the following critical values:

 $P(\bar{W} > 3.23) = 0.10, \quad P(\bar{W} > 4.26) = 0.05, \quad P(\bar{W} > 6.81) = 0.01.$

This means that we reject the null hypothesis at the 10% significance level if

$$\int \mathcal{S}\left(z_N(\gamma)z_N'(\gamma)\right)d\gamma / \int \mathcal{S}\left(\sigma_N(\gamma,\gamma)\right)d\gamma > 3.23\tag{30}$$

To prove Theorem 1 we need the following definitions and assumption. *Definitions:*

$$V_{11} = E(X'_i X_i) \qquad \qquad \Omega_{11} = E(X'_i X_i \varepsilon_i^2) \\ V_{12}(\gamma) = E(X'_i Z_i | I(q_i = \gamma)) \qquad \qquad \Omega_{12}(\gamma) = E(X'_i Z_i \varepsilon_i^2 | I(q_i = \gamma)) \\ V_{21}(\gamma) = E(Z'_i X_i | I(q_i = \gamma)) \qquad \qquad \Omega_{21}(\gamma) = E(Z'_i X_i \varepsilon_i^2 | I(q_i = \gamma)) \\ V_{22}(\gamma) = E(Z'_i Z_i | I(q_i = \gamma)) \qquad \qquad \Omega_{22}(\gamma) = E(Z'_i Z_i \varepsilon_i^2 | I(q_i = \gamma))$$

and

$$V(\gamma) = \begin{bmatrix} V_{11} & V_{12}(\gamma) \\ V_{21}(\gamma) & V_{22}(\gamma) \end{bmatrix}, \quad \Omega(\gamma) = \begin{bmatrix} \Omega_{11} & \Omega_{12}(\gamma) \\ \Omega_{21}(\gamma) & \Omega_{22}(\gamma) \end{bmatrix}$$

We also need the s - th moment of these conditional variance matrices:

$$\begin{array}{ll} V_{11}^{s} = E(X_{i}'X_{i})^{s} & \Omega_{11}^{s} = E(X_{i}'X_{i}\varepsilon_{i}^{2})^{s} \\ V_{12}^{s}(\gamma) = E((X_{i}'Z_{i})^{s}|I(q_{i}=\gamma)) & \Omega_{12}^{s}(\gamma) = E((X_{i}'Z_{i}\varepsilon_{i}^{2})^{s}|I(q_{i}=\gamma)) \\ V_{21}^{s}(\gamma) = E((Z_{i}'X_{i})^{s}|I(q_{i}=\gamma)) & \Omega_{21}^{s}(\gamma) = E((Z_{i}'X_{i}\varepsilon_{i}^{2})^{s}|I(q_{i}=\gamma)) \\ V_{22}^{s}(\gamma) = E((Z_{i}'Z_{i})^{s}|I(q_{i}=\gamma)) & \Omega_{22}^{s}(\gamma) = E((Z_{i}'Z_{i}\varepsilon_{i}^{2})^{s}|I(q_{i}=\gamma)) \\ \end{array}$$

and

$$V^{s}(\gamma) = \begin{bmatrix} V_{11}^{s} & V_{12}^{s}(\gamma) \\ V_{21}^{s}(\gamma) & V_{22}^{s}(\gamma) \end{bmatrix}, \quad \Omega^{s}(\gamma) = \begin{bmatrix} \Omega_{11}^{s} & \Omega_{12}^{s}(\gamma) \\ \Omega_{21}^{s}(\gamma) & \Omega_{22}^{s}(\gamma) \end{bmatrix}$$

Finally, we denote f(q) as the probability density of the variable q. Assumptions:¹³

1. Let the series $(X_i, q_i, \varepsilon_i)$ be strictly stationary with β mixing coefficients β_m satisfying

$$\beta_m^{(s-1)/2s} = O(m^{-(1+\delta)}).$$

- 2. $E(\varepsilon_i | F_{i-1}) = 0;$
- 3. $E|X_i|^{2s} < \infty$, and $E|\varepsilon_i|^{2s} < \infty$;
- 4. $f(\gamma), V(\gamma), \Omega(\gamma), V^s(\gamma), \Omega^s(\gamma)$ be continuous at $\gamma = \gamma_0$;
- 5. $V(\gamma), \Omega(\gamma)$ are positive definite and $f(\gamma) > 0$;
- 6. $P(q_i \in \Gamma) < 1.$

Under these assumptions the following lemma can be stated:

¹³These assumptions are similar to those made by Hansen [22], [23]. For a discussion, see Hansen [23].

Lemma 6 $\frac{1}{\sqrt{N}}Z'_{(\gamma)}\varepsilon \Longrightarrow G(\gamma)$, a zero mean Gaussian process with covariance kernel $\sigma^2 V_{22}(\gamma)$, where $V_{22}(\gamma)$ is defined as above and $\gamma = \min(\gamma_1, \gamma_2)$.

The proof of the lemma follows directly from Theorem 3 of Hansen [22] and the above definitions. We are now ready to prove Theorem 1.

Proof of Theorem 1: Define $X^*_{(\gamma)} = [X \ Z_{(\gamma)}]$. Then, for fixed γ ,

$$\frac{1}{\sqrt{N}}X_{(\gamma)}^{*}\varepsilon = \begin{bmatrix} \frac{1}{\sqrt{N}}X'\varepsilon\\ \frac{1}{\sqrt{N}}Z_{(\gamma)}'\varepsilon \end{bmatrix} \to N\left(\begin{pmatrix} 0\\ 0 \end{pmatrix}, \sigma^{2}\begin{bmatrix} V & V_{12}(\gamma)\\ V_{21}(\gamma) & V_{22}(\gamma) \end{bmatrix}\right)$$

and also weak convergence

$$\frac{1}{\sqrt{N}}X^*_{(\gamma)}\varepsilon = \left[\begin{array}{c} \frac{1}{\sqrt{N}}X'\varepsilon\\ \frac{1}{\sqrt{N}}Z'_{(\gamma)}\varepsilon\end{array}\right] \Longrightarrow \left[\begin{array}{c} N(0,\sigma^2V)\\ G(\gamma)\end{array}\right],$$

where $G(\gamma)$ is defined as in Lemma 6. Since

$$\begin{aligned} \frac{1}{\sqrt{N}} Z'_{(\gamma)} M \varepsilon &= \frac{1}{\sqrt{N}} Z'_{(\gamma)} (I - X(X'X)^{-1}X') \varepsilon \\ &= \frac{1}{\sqrt{N}} Z'_{(\gamma)} \varepsilon - \left(\frac{1}{N} Z'_{(\gamma)} X\right) \left(\frac{1}{N} X'X\right)^{-1} \left(\frac{1}{\sqrt{N}} X'\varepsilon\right) \\ &= \left[-\left(\frac{1}{N} Z'_{(\gamma)} X\right) \left(\frac{1}{N} X'X\right)^{-1} I \right] \frac{1}{\sqrt{N}} X^*_{(\gamma)} \varepsilon \\ &\Longrightarrow G^*(\gamma) \end{aligned}$$

where $G^*(\gamma)$ is a zero mean Gaussian process with covariance kernel

$$- ((V_{21}) (V_{11})^{-1}) (\sigma^2 V_{11}) ((V_{21}) (V_{11})^{-1})' + \sigma^2 V_{22}(\gamma)$$

$$= \sigma^2 (-Z'_{(\gamma)} X (X'X)^{-1} X' Z_{(\gamma)} + Z'_{(\gamma)} Z_{(\gamma)})$$

$$= \sigma^2 Z'_{(\gamma)} M Z_{(\gamma)}.$$

The theorem follows. ■

No.	Author(s)	Title
9773	H. Huizinga and S.B. Nielsen	The Taxation of Interest in Europe: A Minimum Withholding Tax?
9774	E. Charlier	Equivalence Scales for the Former West Germany
9775	M. Berliant and T. ten Raa	Increasing Returns and Perfect Competition: The Role of Land
9776	A. Kalwij, R. Alessie and P. Fontein	Household Commodity Demand and Demographics in the Netherlands: a Microeconometric Analysis
9777	P.J.J. Herings	Two Simple Proofs of the Feasibility of the Linear Tracing Procedure
9778	G. Gürkan, A.Y. Özge and S.M. Robinson	Sample-Path Solutions for Simulation Optimization Problems and Stochastic Variational Inequalities
9779	S. Smulders	Should Environmental Standards be Tighter if Technological Change is Endogenous?
9780	B.J. Heijdra and L. Meijdam	Public Investment in a Small Open Economy
978 <mark>1</mark>	E.G.F. Stancanelli	Do the Rich Stay Unemployed Longer? An Empirical Study for the UK
9782	J.C. Engwerda and R.C. Douven	Local Strong <i>d</i> -Monotonicity of the Kalai-Smorodinsky and Nash Bargaining Solution
9783	J.C. Engwerda	Computational Aspects of the Open-Loop Nash Equilibrium in Linear Quadratic Games
9784	J.C. Engwerda, B. van Aarle J.E.J. Plasmans	The (In)Finite Horizon Open-Loop Nash LQ-Game: An Application to EMU
9785	J. Osiewalski, G. Koop and M.F.J. Steel	A Stochastic Frontier Analysis of Output Level and Growth in Poland and Western Economies
9786	F. de Jong	Time-Series and Cross-Section Information in Affine Term Structure Models
9787	G. Gürkan, A.Y. Özge and S.M. Robinson	Sample-Path Solution of Stochastic Variational Inequalities
9788	A.N. Banerjee	Sensitivity of Univariate AR(1) Time-Series Forecasts Near the Unit Root
9789	G. Brennan, W. Güth and H. Kliemt	Trust in the Shadow of the Courts
9790	A.N. Banerjee and J.R. Magnus	On the Sensitivity of the usual <i>t</i> - and <i>F</i> -tests to AR(1) misspecification
9791	A. Cukierman and M. Tommasi	When does it take a Nixon to go to China?

No.	Author(s)	Title
9792	A. Cukierman, P. Rodriguez and S.B. Webb	Central Bank Autonomy and Exchange Rate Regimes - Their Effects on Monetary Accommodation and Activism
9793	B.G.C. Dellaert, M. Prodigalidad and J.J. Louvriere	Family Members' Projections of Each Other's Preference and Influence: A Two-Stage Conjoint Approach
9794	B. Dellaert, T. Arentze, M. Bierlaire, A. Borgers and H. Timmermans	Investigating Consumers' Tendency to Combine Multiple Shopping Purposes and Destinations
9795	A. Belke and D. Gros	Estimating the Costs and Benefits of EMU: The Impact of External Shocks on Labour Markets
9796	H. Daniëls, B. Kamp and W. Verkooijen	Application of Neural Networks to House Pricing and Bond Rating
9797	G. Gürkan	Simulation Optimization of Buffer Allocations in Production Lines with Unreliable Machines
9798	V. Bhaskar and E. van Damme	Moral Hazard and Private Monitoring
9799	F. Palomino	Relative Performance Equilibrium in Financial Markets
97100	G. Gürkan and A.Y. Özge	Functional Properties of Throughput in Tandem Lines with Unreliable Servers and Finite Buffers
97101	E.G.A. Gaury, J.P.C. Kleijnen and H. Pierreval	Configuring a Pull Production-Control Strategy Through a Generic Model
97102	F.A. de Roon, Th.E. Nijman and C. Veld	Analyzing Specification Errors in Models for Futures Risk Premia with Hedging Pressure
97103	M. Berg, R. Brekelmans and A. De Waegenaere	Budget Setting Strategies for the Company's Divisions
97104	C. Fernández and M.F.J. Steel	Reference Priors for Non-Normal Two-Sample Problems
97105	C. Fernández and M.F.J. Steel	Reference Priors for the General Location-Scale Model
97106	M.C.W. Janssen and E. Maasland	On the Unique D1 Equilibrium in the Stackelberg Model with asymmetric information
9 7 107	A. Belke and M. Göcke	Multiple Equilibria in German Employment -Simultaneous Identification of Structural Breaks-
97108	D. Bergemann and U. Hege	Venture Capital Financing, Moral Hazard, and Learning
97109	U. Hege and P. Viala	Contentious Contracts
97110	P.JJ. Herings	A Note on "Stability of Tâtonnement Processes of Short Period Equilibria with Rational Expectations"

No.	Author(s)	Title
9 7 111	C. Fernández, E. Ley, and M.F.J. Steel	Statistical Modeling of Fishing Activities in the North Atlantic
97112	J.J.A. Moors	A Critical Evaluation of Mangat's Two-Step Procedure in Randomized Response
97113	J.J.A. Moors, B.B. van der Genugten, and L.W.G. Strijbosch	Repeated Audit Controls
97114	X. Gong and A. van Soest	Family Structure and Female Labour Supply in Mexico City
97115	A. Blume, D.V. DeJong, YG. Kim and G.B. Sprinkle	Evolution of Communication with Partial Common Interest
97116	J.P.C. Kleijnen and R.G. Sargent	A Methodology for Fitting and Validating Metamodels in Simulation
97117	J. Boone	Technological Progress and Unemployment
97118	A. Prat	Campaign Advertising and Voter Welfare
9801	H. Gersbach and H. Uhlig	Debt Contracts, Collapse and Regulation as Competition Phenomena
9802	P. Peretto and S. Smulders	Specialization, Knowledge Dilution, and Scale Effects in an IO- based Growth Model
9803	K.J.M. Huisman and P.M. Kort	A Further Analysis on Strategic Timing of Adoption of New Technologies under Uncertainty
9804	P.JJ. Herings and A. van den Elzen	Computation of the Nash Equilibrium Selected by the Tracing Procedure in <i>N</i> -Person Games
9805	P.JJ. Herings and J.H. Drèze	Continua of Underemployment Equilibria
9806	M. Koster	Multi-Service Serial Cost Sharing: A Characterization of the Moulin-Shenker Rule
9807	F.A. de Roon, Th.E. Nijman and B.J.M. Werker	Testing for Mean-Variance Spanning with Short Sales Constraints and Transaction Costs: The Case of Emerging Markets
9808	R.M.W.J. Beetsma and P.C. Schotman	Measuring Risk Attitudes in a Natural Experiment: Data from the Television Game Show Lingo
9809	M. Bütler	The Choice between Pension Reform Options
9810	L. Bettendorf and F. Verboven	Competition on the Dutch Coffee Market
9811	E. Schaling, M. Hoeberichts and S. Eijffinger	Incentive Contracts for Central Bankers under Uncertainty: Walsh-Svensson non-Equivalence Revisited

No.	Author(s)	Title
9812	M. Slikker	Average Convexity in Communication Situations
9813	T. van de Klundert and S. Smulders	Capital Mobility and Catching Up in a Two-Country, Two-Sector Model of Endogenous Growth
9814	A.Belke and D. Gros	Evidence on the Costs of Intra-European Exchange Rate Variability
9815	J.P.C. Kleijnen and O. Pala	Maximizing the Simulation Output: a Competition
9816	C. Dustmann, N. Rajah and A. van Soest	School Quality, Exam Performance, and Career Choice
9817	H. Hamers, F. Klijn and J. Suijs	On the Balancedness of <i>m</i> -Sequencing Games
9818	S.J. Koopman and J. Durbin	Fast Filtering and Smoothing for Multivariate State Space Models
9819	E. Droste, M. Kosfeld and M. Voorneveld	Regret Equilibria in Games
9820	M Slikker	A Note on Link Formation
9821	M. Koster, E. Molina, Y. Sprumont and S. Tijs	Core Representations of the Standard Fixed Tree Game
9822	J.P.C. Kleijnen	Validation of Simulation, With and Without Real Data
9823	M. Kosfeld	Rumours and Markets
9824	F. Karaesmen, F. van der Duyn Schouten and L.N. van Wassen- hove	Dedication versus Flexibility in Field Service Operations
9825	J. Suijs, A. De Waegenaere and P. Borm	Optimal Design of Pension Funds: A Mission Impossible
9826	U.Gneezy and W. Güth	On Competing Rewards Standards -An Experimental Study of Ultimatum Bargaining-
9827	M. Dufwenberg and U. Gneezy	Price Competition and Market Concentration: An Experimental Study
9828	A. Blume, D.V. De Jong and G.R. Neumann	Learning in Sender-Receiver Games
9829	B.G.C. Dellaert, J.D. Brazell and J.J. Louviere	Variations in Consumer Choice Consistency: The Case of Attribute-Level Driven Shifts in Consistency
9830	B.G.C. Dellaert, A.W.J. Borgers, J.J. Louviere and H.J.P. Timmermans	Consumer Choice of Modularized Products: A Conjoint choice Experiment Approach

No.	Author(s)	Title
9831	E.G.A. Gaury, H. Pierreval and J.P.C. Kleijnen	New Species of Hybrid Pull Systems
9832	S.J. Koopman and H.N. Lai	Modelling Bid-Ask Spreads in Competitive Dealership Markets
9833	F. Klijn, M. Slikker, S. Tijs and J. Zarzuelo	Characterizations of the Egalitarian Solution for Convex Games
9834	C. Fershtman, N. Gandal and S. Markovich	Estimating the Effect of Tax Reform in Differentiated Product Oligopolistic Markets
9835	M. Zeelenberg, W.W. van Dijk, J. van der Pligt, A.S.R. Manstead, P. van Empelen and D. Reinderman	Emotional Reactions to the Outcomes of Decisions: The Role of Counterfactual Thought in the Experience of Regret and Disappointment
9836	M. Zeelenberg, W.W. van Dijk and A.S.R. Manstead	Reconsidering the Relation between Regret and Responsibility
9837	M. Dufwenberg and G. Kirchsteiger	A Theory of Sequential Reciprocity
9838	A. Xepapadeas and A. de Zeeuw	Environmental Policy and Competitiveness: The Porter Hypothesis and the Composition of Capital
9839	M. Lubyova and J.C. van Ours	Unemployment Durations of Job Losers in a Labor Market in Transition
9840	P. Bolton and X. Freixas	A Dilution Cost Approach to Financial Intermediation and Securities Markets
9841	A. Rustichini	Minimizing Regret: The General Case
9842	J. Boone	Competitive Pressure, Selection and Investments in Development and Fundamental Research
9843	H.L.F. de Groot	Macroeconomic Consequences of Outsourcing. An Analysis of Growth, Welfare, and Product Variety
9844	U. Gneezy, W. Güth and F. Verboven	Presents or Investments? An Experimental Analysis
9845	A. Prat	How Homogeneous Should a Team Be?
9846	P. Borm and H. Hamers	A Note on Games Corresponding to Sequencing Situations with Due Dates
9847	A.J. Hoogstrate and T. Osang	Saving, Openness, and Growth

